Author's response to reviews

Title: Benzodiazepine prescribing behaviour and attitudes: a survey among general practitioners practicing in northern Thailand

Authors:

Manit Srisurapanont (msrisura@mail.med.cmu.ac.th)
Paul Garner (pgarner@liverpool.ac.uk)
Julia Critchley (juliac@liverpool.ac.uk)
Nahathai Wongpakaran (nkuntawo@mail.med.cmu.ac.th)

Version: 5  Date: 16 June 2005

Author's response to reviews:

Formatting and wording have been revised (very little) to be in concordance with the journal instruction.